Author’s response to reviews

Title: Identification of factors associated with diagnostic error in primary care

Authors:

Sergio Minue (sergio.minue.easp@juntadeandalucia.es)
Clara Bermudez-Tamayo (clara.bermudez.tamayo@umontreal.ca)
Alberto Fernandez Ajuria (alberto.fernandez.easp@juntadeandalucia.es)
Jose Martin Martin (jmartin@ugr.es)
Vivian Benitez (vivben75@gmail.com)
Miguel Melguizo (oziuglemm@telefonica.net)
Araceli Caro (araceli.caro.easp@juntadeandalucia.es)
Maria Jose Orgaz (mjorgazrosua@icloud.com)
Miguel Angel Prados (mapk3@gmail.com)
Enrique Diaz (enriquediaz01@gmail.com)
Rafael Montoro (rafael.montoro.sspa@juntadeandalucia.es)

Version: 2  Date: 7 April 2014

Author’s response to reviews: see over
April 7th 2014
Escuela Andaluza de Salud Pública

Dear Editor BMC,

Please find enclosed our manuscript: Identification of factors associated with diagnostic error in primary care, that we submit to Implementation Science, to be considered for publication.

- This manuscript is original, and has not been sent to any other journal.
- Each author has contributed substantially to the conception and design of the study, the analysis and interpretation of information. All authors have also approved the final draft of the manuscript and, therefore, they can take responsibility for its content.
- I confirm that I have forwarded copies of ethical approval and funding approval to BMCSeriesEditorial@biomedcentral.com. The study was funded by the Health Ministry of Health (Spain).
- It is an ‘ongoing’ study, the investigators are still collecting, or analyzing data.
- I confirm that there is not yet any publication (published or submitted) containing the results of this study.

The authors have no competing interests to declare.

Yours sincerely,

Sergio Minue
Corresponding Author
e-mail: isabel.ruiz.easp@juntadeandalucia.es